 Vc/29-Mar-04
Week 14  starting 29 March 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for this week : 

	Bench

	Name

	Status Monday 1730

	Temp

	-

	-

	 

	 


	TBB1

	MBAR1051

	Commissioning

	300 K

	 

	 

	 

	 


	TBB2

	MBBL3122

	            "

	300 K

	 

	 

	 

	 


	TBC1

	MBAR2016

	Prep 1 Lyre Test

	300 K

	 

	 

	 

	 


	TBC2

	SSS0013

	PT 7.1 detrain check

	1.9 K

	 

	 

	 

	 


	TBD1

	MBAL1058

	CR 1 Leak Test

	300 K

	 

	 

	Try Keep free After


	TBD2

	MBBL3033

	Prep 5 Pumping Cool Down

	Cooling

	 

	 

	 

	 


	TBE1

	MBBL2063

	PT 6.1 Training 1

	1.9 K

	 

	 

	 

	 


	TBE2

	MBBR3018

	ICS 3  after warm test

	300 K

	 

	 

	 

	 


	TBF1

	1061

	NEW Last night

	300 K

	 

	 

	 

	 


	TBF2

	1055

	NEW Last night

	 

	 

	 

	 

	 



	
	
	
	
	
	

	
	
	
	
	
	


Test Coordinator this week:  Pierre Pugnat; MTM support on call  Federico:  (16-3602).

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
· Keep D1 free for SSS Mech Preparation ( 6 hrs ? )   J.Halik 
· PLC For Water & Federico/Alessandro This week at night ?
	Date
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 23_M


	TBF1: Difference in T magnet (2011). TT821 reading : 5.1 K.

T cryo: 1.9 K 

HV test failed.
	MTM
	
	TT 821 sensor is shorted to the ground.



	Tuesday 23_M


	TBE: TEMA locked (Florence?).
	AB-CO
	TEMA
	Giorgio called and unlocked TEMA

	Tuesday 23_A


	TBF1: Min Energy@11850 A analysis: Peak voltage limit more than 100 V for D2-L. D2-U is flat.
	MTM
	
	One HF acquisition cable replaced by Giorgio (flat curve). Mail reported to P.Pugnat as specialist  (V too high).

	Tuesday 23_A


	TBD2: “Error at CPU address during the encoder pulse reception” during the machine cycle of the 2005.


	AB-CO


	
	On acknowledgment, the error disappeared.



	Wednesday 24_N


	TBE2: Temperature (TE148) sensor problem, which pilots the phases. 


	ACR


	
	Conditioner changed 



	Wednesday 24_N


	TBD2: After the quench for warm up, the vacuum degraded (reached 3 mbar).

Origin not known.

 
	ACR
	
	Cryo team installed a mobile pump but temperature was falling. Vacuum recovered after 3 hours. Giorgio called and advised not to remove shaft as they were already at 200 K.

They were removed after the warm up and re-calibrated.

	Thursday 25_A


	TBD1: HV test at cold: problem with SunMTA36.

 
	AB/CO
	
	Giorgio and Richard called Pedro. 

Test performed on TBE sun.

RS232 transducer changed. 

	Friday 26_A


	TBD2: HV test @ warm done by ICS for the 3033 magnet. Test OK but file not created or not transferred?

 
	MTM/AB-CO
	
	Richard is investigating.

	Friday 26_A


	TBD2: Launching IAP@warm, HWREC error message:” configuration files contain Oracle errors “

Test done with previous LF config file:  Problem with T sensor (263 K).
	MTM
	HWREC
	Giorgio informed. He came and created the proper config file (1 Qloc crate missing from the bench because of TB-B commissioning). Test OK.

	Friday 26_N


	TBF2: PC fault (water) before launching the SPA. Cannot reset the alarm.

 
	AB/PO
	
	Giorgio investigated and discovered a water flow too low internal to the PC (RMF).

OK.

	Sunday 28_A


	TBD2: Configuration with anti-cryostat, flow of N2.But magnet without anti cryosytat.

 
	MTM/ACR??
	
	To be clarified.

	Sunday 28_A


	Calibration bench: Water leak. Valves closed

 
	MTM
	
	Intervention of J.Halik Monday

	Monday 29_M


	TBF1: IAP @warm final .3 faulty files created (magnet 1078)

 
	MTM
	
	Files to be erased.

Stop of acquisition forgotten after IAP?



	Monday 29_M


	TBE1: Training quench stopped,

General problem of pumping.  Warmer H1 down few minutes without alarms. Then it restarted but pressure was rising in the lines.

 
	ACR
	
	Now OK but investigation on the run to understand the origin of the stop.










